Author’s response to reviews

Title: NON-SYNDROMIC AORTIC VALVE MYXOMA IN A TEEN, CAUSE OF ANGINA SYMPTOMS

Authors:

Geoffrey Changwe (drchangwegj1984@gmail.com)
Tao Zhang (zhangt258@126.com)
Zhang Haizhou (zhz_doctor@163.com)
Chengwei Zou (zouchengwei@sdu.edu.cn)

Version: 1 Date: 05 Jul 2019

Author’s response to reviews:

Reviewer #1.
Concern: Availability of morphological study report for differential diagnosis?

Feedback: Your concern is appreciated. Morphological report has been provided for you and readers to view.

Reviewer #2.
Concern: any scanning modality done to exclude coronary artery lesion?

Feedback: Your concern is valid and appreciated. A sentence in the 11th line under CASE PRESENTATION paragraph of the manuscript reads: Other imaging modalities were non-remarkable. Indeed, MRI was done to exclude coronary artery lesions.